Author's response to reviews

Title: Predictors of Oral Rehydration Therapy use among under-five children with diarrhea in Eastern Ethiopia: A community based case control study

Authors:

Bezatu Mengistie Alemu (bezatum@gmail.com)
Yemane Berhane - (yemaneeberhane@addiscontinental.edu.et)
Alemyahu worku - (alemayehuwy@yahoo.com)

Version: 4 Date: 21 November 2012

Author's response to reviews: see over
To: Public Health Journal, Editorial Board

Dear Editor-in-Chief,

Please find the copy edited manuscript entitled "Predictors of Oral Rehydration Therapy use among under-five children with diarrhea in Eastern Ethiopia: community based study."

We apology for the delay. The language editor has taken more time than we expected.

Best regards,

Bezatu Mengistie, also on behalf of the co-authors

Department of Public Health
Haramaya University
P.O.Box 235, Harar
Ethiopia
Email: bezex2000@yahoo.com or bezatum@gmail.com